Author's response to reviews

Title: Cocaine use during pregnancy assessed by hair analysis in a Canary Islands cohort

Authors:

Xavier Joya (jjoya@imim.es)
Mario Gomez-Culebras (mgomez@ull.es)
Alicia Callejon (acallejon@ull.es)
Bibiana Friguls (bfriguls@imim.es)
Carme Puig (carme.puig@yahoo.es)
Sandra Ortigosa (sortigosa@imas.imim.es)
Luca Morini (luca.morini@unipv.it)
Oscar Garcia-Algar (90458@imas.imim.es)
Oriol Vall (ovall@imas.imim.es)

Version: 8 Date: 5 January 2012

Author's response to reviews:

small typographical changes have been made